Author’s response to reviews

Title: A landscape of genomic alterations at the root of a near-untreatable tuberculosis epidemic

Authors:

Marisa Klopper (marisat@sun.ac.za)
Tim Heupink (tim.heupink@uantwerpen.be)
Grant Hill-Cawthorne (grant.hill-cawthorne@sydney.edu.au)
Elizabeth Streicher (lizma@sun.ac.za)
Anzaan Dippenaar (adippenaar@sun.ac.za)
Margaretha de Vos (margaretha.devos@finddx.org)
Abdallah Abdallah (abdallah.musa@qu.edu.qa)
Jason Limberis (jason.limberis@uct.ac.za)
Matthias Merker (mmerker@fz-borstel.de)
Scott Burns (yqd0@cdc.gov)
Stefan Niemann (stniemann@yahoo.de)
Keertan Dheda (keertan.dheda@uct.ac.za)
James Posey (hzp9@cdc.gov)
Arnab Pain (arnab.pain@kaust.edu.sa)
Robin Warren (rw1@sun.ac.za)

Version: 3 Date: 18 Dec 2019

Author’s response to reviews:

Dear Lin

Thank you for reviewing our manuscript, we are thrilled that you are willing to publish it.

I have made the changes as you suggested:
1. I have re-ordered the supplemental material to appear chronologically. However, supplements 8-10 are only mentioned in the "data availability" section (lines 557-559) - does this count as main text, or would you like me to put it in the methods section as well (which would mean re-ordering everything again)?
2. I have re-ordered the Declarations section according to your specifications. (lines 526-576)

3. The email address for Dr Abdallah as you had it (abdallah.musa@qu.edu.qa), is correct according to my records, but I have been communicating with him via his gmail address: dr.abdallahmusa@gmail.com. I have changed his contact email to the gmail address.

Looking forward to your response.

Kind regards

Marisa